Author's response to reviews

Title: Provision of relapse prevention interventions in UK NHS Stop Smoking Services: a survey

Authors:

Shade A Agboola (shade.agboola@nottingham.ac.uk)
Tim J Coleman (tim.coleman@nottingham.ac.uk)
Jo A Leonardi-Bee (jo.leonardi-bee@nottingham.ac.uk)
Andy McEwen (andy.mcewen@ucl.ac.uk)
Ann D McNeill (ann.mcneill@nottingham.ac.uk)

Version: 7 Date: 19 July 2010

Author's response to reviews:

Dear Editorial Production Team,

I have amended the name abbreviations on the first page of the manuscript for Andy McEwen and Ann McNeill to read AM followed by email address for Andy McEwen and AMc for Ann McNeill followed by email address. Thank you.

Sincerely,

Dr Agboola